Author's response to reviews

Title: Immediate vs. Delayed Insertion of Intrauterine Contraception after Second Trimester Abortion: Study Protocol for a Randomized Controlled Trial

Authors:

Wendy V. Norman (wvnorman@interchange.ubc.ca)
Janusz Kaczorowski (janusz.kaczorowski@familymed.ubc.ca)
Judith A Soon (jasoon@interchange.ubc.ca)
Rollin Brant (rollin@stat.ubc.ca)
Stirling Bryan (stirling.bryan@ubc.ca)
Konia J Trouton (konia.trouton@viha.ca)
Lyda Dicus (ldicus@cw.bc.ca)

Version: 7 Date: 9 June 2011

Author's response to reviews: see over
Editors-in-Chief,
BMC Trials,

2011 June 9th

Dear Sirs;

Thank you for reviewing this randomized controlled trial protocol manuscript “Immediate vs. Delayed Insertion of Intrauterine Contraception after Second Trimester Abortion: Protocol and Rationale for a Randomized Controlled Trial ”, and accepting for publication in Biomed Central’s journal “Trials”.

We have corrected the previous error in the citations of references, have provided a title for Table 3 (previously missing) and have reviewed thoroughly for typographical errors.

We thank you for accepting this manuscript for publication, and for any other suggestions you may have.

With kind regards,

\[signature\]

W. V. Norman
MD, FCFP, DTM&H, MHSc,
Clinical Professor,
Faculty of Medicine,
University of British Columbia